Author’s response to reviews

Title: Assessing the risk factors before pregnancy of preterm births in Iran: a population-based case-control study

Authors:
Maryam Soltani (maryam.soltani52@yahoo.com)
Hamid Reza Tabatabaee (tabatabaee@sums.ac.ir)
Shahin Saeidinejat (saeedish1@mums.ac.ir)
Marzieh Eslahi (marzieh.eslahi@gmail.com)
Halime Yaghoobi (Lemen_2012@yahoo.com)
Ehsan Mazloumi (Ehsan.mazloumi1@gmail.com)
Abdolhalim Rajabi (rajabiepid@gmail.com)
Ali Ghasemi (aria_201323@yahoo.com)
Naeimeh Keyghobadi (naeimehkeyghobady@yahoo.com)
Mostafa Enayatrad (mostafaenayatrad@gmail.com)
Abed Noori (Abed.noori@gmail.com)
Seyyed Aliasghar Hashemi (hashemishiz@gmail.com)
Fatemeh Zolfizadeh (fa.9712@yahoo.com)
Sepideh Mahdavi (S.mahdavi88@gmail.com)
Tannaz Valadbeigi (Tanazvaladbeigi@gmail.com)
Koorosh Etemad (etemadk@gmail.com)
Ali Taghipour (Taghipour1@mums.ac.ir)
Cirruse Salehnasab (Salehnasab@yahoo.com)
Mahmoud Hajipour (m.hajipour.13@gmail.com)

Version: 7 Date: 05 Jan 2019
Author’s response to reviews:

Dear Editor,

Thank you for your attention. According to your comment "Please note that your form has now been approved, but we note that CS (Cirruse Salehnasab) is not included in the Authors contributions section" we have changed SS (Cirruse Salehnasab) to CS (Cirruse Salehnasab) in Authors contributions section of revised manuscript.

Sincerely,

Tannaz Valabeigi